Author's response to reviews

Title: Prevalence of cardiovascular risk factors among adults without obvious cardiovascular disease in a rural community in Ekiti State, Southwest Nigeria

Authors:

Olarinde Jeffery Ogunmola (joogunmola@yahoo.com)
Adeleke Oluwasegun Olaifa (chemiesweet44@yahoo.com)
Olu Lola Oladapo (loladapo2000@yahoo.co.uk)
Oluwole Adeyemi Babatunde (wolleking@hotmail.com)

Version: 6 Date: 4 October 2013

Author's response to reviews: see over
RESPONSE TO REVIEWER’S REPORT

REVIEWER 1 [Dr Familoni]

Reviewer's report:
I am satisfied with the corrections and explanations made and recommend that manuscript be accepted as amended

Response: I hereby tender my sincere appreciation and thanks on his contribution to improving the quality of the paper.

RESPONSE TO REVIEWER’S REPORT

REVIEWER 2 [Michael O. Balogun]

Reviewer's report:
Major Compulsory Revisions: None
Minor Essential Revisions: Typographical error under - Ethical considerations page 7 line 3: The value of waist circumference should have the correct symbol in front of it.

Response: Sir, i need a clarification concerning your comment-
"The value of waist circumference should have the correct symbol in front of it". I presume possibly you are referring to the unit which i had corrected. See attachment for the copy i sent for your view sir on page 7 line 3.